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Abstract 

Fructan is a polymer of fructose and can be categorised into various types, such as 

inulin, levan, and agavin, and graminan, based on their linkage patterns and degree of 

polymerization. Owing to the physiological effects induced, these have been used as prebiotics, 

and in foods and beverages, pharmaceutical, and cosmetic industries. With increasing interest 

in both high molecular weight (HMW) and short-chain fructans, more efficient methods need 

to be developed for their production. In this study, we disclose the biosynthesis of high 

molecular weight (HMW) levan employing whole cells of Microbacterium paraoxydans. The 

structural characterization of the fructan produced extracellularly revealed predominant β-2,6 

linkages between adjacent fructose units. Notably, over 90% of the fructan exhibited a 

molecular weight of approximately 2 x 103 kDa, while the remaining 10% displayed a lower 

molecular weight of around 20 kDa. The optimization of factors influencing the conversion 

rate was conducted through a systematic One-Factor-At-a-Time (OFAT) analysis. The 

synergistic effects of these optimized conditions, namely an initial sucrose concentration of 

400 g/L, 100 mM MOPS buffer (pH 7), a temperature of 37 °C, and the addition of 20 mM 

CaCl2, resulted in the production of approximately 129 g/L of levan. The achieved yield was 

approximately 0.41 g/g sucrose consumed, against the theoretical yield of 0.47 g/g, 

accompanied by a volumetric productivity of 1.8 g/L/h. 

For the production of 1-kestose enriched short chain-fructooligosaccharides (scFOS), a 

plant-derived recombinant fructosyltransferase was explored. High-level extracellular 

production of the recombinant sucrose: sucrose 1-fructosyltransferase (r1-SST, EC 2.4.1.99) 

from Festuca arundinacea was achieved in Komagataella phaffii (formerly Pichia pastoris) 

X-33 strain containing 3 copies of the codon-optimized 1-SST gene. Continuous production of 

the r1-SST was carried out in a 2 L bioreactor. In a bid to overcome the redox imbalance and 
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hypoxic stress caused in bioreactor cultivation due to inadequate mixing and methanol uptake, 

a single-run chemostat-based strain characterization methodology was used to evaluate the 

effect of methanol-sorbitol co-feeding strategy and the specific growth rate (µ) on the cell-

specific r1-SST productivity (Qp) and the cell-specific oxygen uptake rate (Qo). The co-

feeding showed ~46% lower Qo for similar Qp than the methanol-alone feeding strategy. At 

an optimum dilution rate of 0.025 h-1, a maximum Qp value of ~100 U/g dry cell weight/h was 

achieved at a steady-state biomass concentration of ~15 g/L. Finally, high-cell-density 

continuous cultivation demonstrated continuous production of the r1-SST for a sustained 

period of 15 days at 0.025 h-1 achieving 78% of the maximum Qp value at a steady-state 

biomass concentration of ~64 g/L and volumetric productivity of ~5000 U/L/h. A kinetic model 

based on a combined approach of rapid equilibrium and steady-state kinetics was developed to 

predict the time profile of production of scFOS using purified r1-SST. The purified enzyme 

was shown to selectively convert sucrose into 1-kestose with a final yield of 0.66 g/g sucrose 

consumed. The model could predict the effect of substrate concentration on initial rate kinetics, 

half-saturation constant, and transferase to hydrolase ratio. For the known values of initial 

sucrose and enzyme concentration, the optimum reaction time and the required dilution rate 

were predicted for the development of a continuous process to produce 1-kestose enriched 

scFOS. A volumetric productivity of ~110 g/L/h of 1-kestose with 40% carbohydrate 

composition in the permeate flux was demonstrated. 
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सार 

 फु्रके्टन फु्रक्टोज का एक बहुलक है और इसे उनके ललिंकेज पैटनन और पोलीमराइजेशन की 

लिग्री के आधार पर लिलिन्न प्रकारोिं में िर्गीकृत लकया जा सकता है, जैसे लक इनुललन, लेिन, और एरे्गलिन 

और गै्रलमनन। पे्रररत शारीररक प्रिािोिं के कारण, इनका उपयोर्ग प्रीबायोलटक्स के रूप में और खाद्य एििं 

पेय पदार्न, फामानसु्यलटकल और कॉसे्मलटक उद्योर्गोिं में लकया र्गया है। उच्च आणलिक िार (एचएमिबू्ल्य) 

और लघु-शृ्िंखला फु्रके्टन दोनोिं में बढ़ती रुलच के सार्, उनके उत्पादन के ललए अलधक कुशल तरीकोिं को 

लिकलसत करने की आिश्यकता है। इस अध्ययन में, हम माइक्रोबैक्टीररयम पैराऑक्सक्सिैन्स की सिंपूणन 

कोलशकाओिं को लनयोलजत करके उच्च आणलिक िार (एचएमिबू्ल्य) लेिन के जैिसिंशे्लषण का खुलासा 

करते हैं। बाह्यकोलशकीय रूप से लनलमनत फु्रके्टन के सिंरचनात्मक लक्षण िणनन से आसन्न फु्रक्टोज इकाइयोिं 

के बीच प्रमुख β-2,6 सिंबिंधोिं का पता चला। लिशेष रूप से, 90% से अलधक फु्रके्टन ने लर्गिर्ग 2 x 103 

kDa का आणलिक िार प्रदलशनत लकया, जबलक शेष 10% ने लर्गिर्ग 20 kDa का आणलिक िार प्रदलशनत 

लकया। रूपािंतरण दर को प्रिालित करने िाले कारकोिं का अनुकूलन एक व्यिक्सित िन-फैक्टर-ए-टाइम 

(ओएफएटी) लिशे्लषण के माध्यम से लकया र्गया र्ा। इन अनुकूललत क्सिलतयोिं के सहलक्रयात्मक प्रिाि, 

अर्ानत् 400 ग्राम/लीटर की प्रारिं लिक सुक्रोज सािंद्रता, 100 mM एमओपीएस बफर (पीएच 7), 37 लिग्री 

सेक्सियस का तापमान, और 20 mM CaCl2 को जोड़ने के पररणामस्वरूप लर्गिर्ग 129 का उत्पादन 

हुआ। लेिन का जी/एल. प्राप्त उपज लर्गिर्ग 0.41 ग्राम/ग्राम सुक्रोज की खपत र्ी, जबलक सैद्ािंलतक 

उपज 0.47 ग्राम/ग्राम र्ी, सार् ही िॉलू्यमेलटिक उत्पादकता 1.8 ग्राम/लीटर/घिंटा र्ी। 

1-केस्टोज़ समृद् लघु शृ्िंखला-फु्रकु्टललर्गोसेकेराइि्स (एससीएफओएस) के उत्पादन के ललए, 

एक पौधे-वु्यत्पन्न पुनः  सिंयोजक फु्रक्टोलसलटि ािंसफेरेज़ का पता लर्गाया र्गया र्ा। पुनः  सिंयोजक सुक्रोज का 

उच्च-स्तरीय बाह्यकोलशकीय उत्पादन: फेसु्टका अरुिं लिनेलसया से सुक्रोज 1-फु्रक्टोलसलटि ािंसफेरेज़ 

(आर1-एसएसटी, ईसी 2.4.1.99) कोमार्गाटेला फाफी (पूिन में लपलचया पास्टोररस) एक्स-33 स्टि ेन में प्राप्त 
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लकया र्गया र्ा लजसमें कोिन-अनुकूललत 1 की 3 प्रलतयािं शालमल र्ी िं। -एसएसटी जीन. आर1-एसएसटी 

का लनरिंतर उत्पादन 2 एल बायोररएक्टर में लकया र्गया। अपयानप्त लमश्ण और मेर्नॉल ग्रहण के कारण 

बायोररएक्टर खेती में होने िाले रेिॉक्स असिंतुलन और हाइपोक्सक्सक तनाि को दूर करने के ललए, 

मेर्नॉल-सोलबनटोल सह-फीलििंर्ग रणनीलत और लिलशष्ट के प्रिाि का मूल्यािंकन करने के ललए एकल-रन 

केमोसे्टट-आधाररत तनाि लक्षण िणनन पद्लत का उपयोर्ग लकया र्गया र्ा। कोलशका-लिलशष्ट r1-SST 

उत्पादकता (Qp) और कोलशका-लिलशष्ट ऑक्सीजन ग्रहण दर (Qo) पर िृक्सद् दर (µ)। मेर्नॉल-अलोन 

फीलििंर्ग रणनीलत की तुलना में सह-फीलििंर्ग ने समान कू्यपी के ललए ~46% कम कू्यओ लदखाया। 0.025 

h-1 की इष्टतम कमजोर पड़ने की दर पर, ~15 g/L की क्सिर-अििा बायोमास सािंद्रता पर ~100 U/g 

शुष्क सेल िजन/h का अलधकतम Qp मान प्राप्त लकया र्गया र्ा। अिंत में, उच्च-कोलशका-घनत्व लनरिंतर 

खेती ने 0.025 एच-1 पर 15 लदनोिं की लनरिंतर अिलध के ललए आर1-एसएसटी के लनरिंतर उत्पादन का 

प्रदशनन लकया, लजससे ~64 ग्राम/लीटर की क्सिर-अििा बायोमास एकाग्रता पर अलधकतम कू्यपी मूल्य 

का 78% प्राप्त हुआ। और ~5000 यू/एल/घिंटा की िॉलू्यमेलटिक उत्पादकता। शुद् आर1-एसएसटी का 

उपयोर्ग करके एससीएफओएस के उत्पादन की समय प्रोफाइल की िलिष्यिाणी करने के ललए तीव्र 

सिंतुलन और क्सिर-अििा कैनेटीक्स के सिंयुक्त दृलष्टकोण पर आधाररत एक र्गलतज मॉिल लिकलसत 

लकया र्गया र्ा। शुद् लकए र्गए एिं जाइम को सुक्रोज को चुलनिंदा रूप से 1-केस्टोस में पररिलतनत करने के 

ललए लदखाया र्गया र्ा, लजसकी अिंलतम उपज 0.66 ग्राम/ग्राम सुक्रोज की खपत र्ी। मॉिल प्रारिं लिक दर 

कैनेटीक्स, अधन-सिंतृक्सप्त क्सिरािंक, और टि ािंसफरेज़ से हाइिि ॉलेज़ अनुपात पर सब्सटि ेट एकाग्रता के प्रिाि 

की िलिष्यिाणी कर सकता है। प्रारिं लिक सुक्रोज और एिं जाइम एकाग्रता के ज्ञात मूल्योिं के ललए, 1-

केस्टोस समृद् एससीएफओएस का उत्पादन करने के ललए एक सतत प्रलक्रया के लिकास के ललए इष्टतम 

प्रलतलक्रया समय और आिश्यक कमजोर पड़ने की दर की िलिष्यिाणी की र्गई र्ी। पमेट फ़्लक्स में 

40% काबोहाइिि ेट सिंरचना के सार् 1-केस्टोज़ की ~110 ग्राम/लीटर/घिंटा की िॉलू्यमेलटिक उत्पादकता 

प्रदलशनत की र्गई। 
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